Author's response to reviews

Title: An unusually large benign tumour (myofibroblastoma) of male breast: a case report

Authors:

AMS Abeysekara (abey@hotmail.co.uk)
HP Priyantha Siriwardana (hppsin@hotmail.com)
KF Abbas (kfabbas@hotmail.co.uk)
Peter Tanner (peter.tanner@bhrhospitals.nhs.uk)
Akin A Ojo (akin.ojo@bhrhospitals.nhs.uk)

Version: 4 Date: 4 December 2007

Author's response to reviews:

The JMCR Editorial Team

Thank you for your e-mail with referees's comments.

We have resubmitted the manuscript with minor corrections suggested by the referees.

1) The word 'unexpected' was included instead of 'unusual' in the text where it is appropriate.

2) A representative microscopic picture of a section of the lesion is submitted as Figure 2.

3) Stains for oestrogen and androgen were not done and a sentence is added in the text to mention that.

Thanking you

yours sincerely

AMS Abeysekara MSc FRCS

Corresponding author